Author’s response to reviews

Title: The effect of massage on localized lumbar muscle fatigue

Authors:

Tim H Tanaka (ttanaka@sympatico.ca)
Gerry Leisman (drgersh@yahoo.com)
Hidetoshi Mori (hmori@k.tsukuba-tech.ac.jp)
Kazushi Nishijo (nishijo@a.tsukuba-tech.ac.jp)

Version: 5 Date: 13 Oct 2002

1. Unnecessary capitalization has been removed from the title.

2. All figures are cropped as closely as possible.

3. The definitions of ICC values have been inserted in the Results section as suggested by Reviewer 1.